Author’s response to reviews

Title: "Crying without tears" in infancy. An early diagnostic sign-post of triple A (Allgrove) syndrome: two case reports

Authors:

Daniel Tibussek (daniel.tibussek@gmx.net)
Sujal Ghosh (sujal.ghosh@med.uni-duesseldorf.de)
Angela Huebner (angela.Huebner@uniklinikum-dresden.de)
Joerg Schaper (joerg.Schaper@med.uni-duesseldorf.de)
Ertan Mayatepek (mayatepek@med.uni-duesseldorf.de)
Katrin Koehler (katrin.koehler@uniklinikum-dresden.de)

Version: 1 Date: 21 Jan 2016

Author’s response to reviews:

Dear reviewers,

thank you for your response to our submission.

As requested we have now added the following sentence to the manuscript (introduction section, last paragraph): "Written informed consent was obtained from the parents of both patients for publication of this case report and the accompanying images"

This sentence was already included to our main document (page 3) in the first submission. But we understand that it has to be included in the actual manuscript.
We have also uploaded the English version of the CARE document.

yours sincerely

Daniel Tibussek